Author's response to reviews

Title: Cytoreductive surgery followed by chemotherapy versus chemotherapy alone for recurrent platinum-sensitive epithelial ovarian cancer (SOCceR trial): a multicenter randomised controlled study

Authors:

rafli van de Laar (r.vandelaar@obgyn.umcn.nl)
Petra LM Zusterzeel (p.zusterzeel@obgyn.umcn.nl)
Toon Van Gorp (toon.van.gorp@mumc.nl)
Marrije R Buist (m.r.buist@amc.uva.nl)
Willemien J van Driel (w.v.driel@nki.nl)
Katja N Gaarenstroom (k.n.gaarenstroom@lumc.nl)
Henriette JG Arts (h.j.g.arts@og.umcg.nl)
Johannes CM van Huisseling (hans.van.huisseling@ghz.nl)
Ralph HM Hermans (ralphhermans@mac.com)
Johanna MA Pijnenborg (h.pijnenborg@tsz.nl)
Eltjo MJ Schutter (e.schutter@mst.nl)
Harold MP Pelikan (h.pelikan@live.nl)
Jos HA Vollebergh (j.vollebergh@bernhoven.nl)
Mirjam JA Engelen (m.engelen@atriummc.nl)
Joanna IntHout (j.inthout@ebh.umcn.nl)
Roy FPM Kruitwagen (r.kruitwagen@mumc.nl)
Leon FAG Massuger (l.massuger@obgyn.umcn.nl)

Version: 2 Date: 5 July 2013

Author's response to reviews: see over
Dear Editor,

We are pleased to resubmit the manuscript of our Study Protocol entitled: ‘Cytoreductive surgery followed by chemotherapy versus chemotherapy alone for recurrent platinum-sensitive epithelial ovarian cancer (SOCceR trial): a multicenter randomised controlled study’. We hope you will find the manuscript suitable for publication in BMC Cancer.

(1) Additional files consisting of the ethical approvals of all participating centers are removed
(2) Acknowledgements are added; the study has no funding
(3) The Ethics paragraph is rephrased; all names of the local ethics committees who already approved the study are mentioned

We look forward to hearing from you on the outcome of the review process.

Yours faithfully,

Rafli van de Laar, MD
On behalf of all co-authors